Author's response to reviews

Title: Evaluation of an inter-professional workshop to develop a psychosocial assessment and child-centred communication training programme for paediatricians in training

Authors:

Debra Nestel (d.nestel@imperial.ac.uk)
Sharon Taylor (sharontaylor_2000@yahoo.com)
Quentin Spender (quentin.spender@btinternet.com)

Version: 4 Date: 17 November 2004

Author's response to reviews:

Affiliations - Italics and authors' job removed
Square bracketed references inserted
Full stop at end of first paragraph on p 17
Conclusion changed to Conclusions
Competing Interests to Competing interests
& removed from reference list
Correctly uploaded figures as outlined
Table border included
Italics removed from table and figure headings
The 2 reviewer's report changes to reference list made